Lawrence Berkeley National Laboratory

LBL Publications

Title

Climate Change Will Increase Biomass Proportion of Global Forest Carbon Stocks Under an SSP5–8.5 Climate Trajectory

Permalink https://escholarship.org/uc/item/3rm748vm

Journal Geophysical Research Letters, 50(23)

ISSN

0094-8276

Authors

Mekonnen, Zelalem A Riley, William J

Publication Date 2023-12-16

DOI 10.1029/2023gl104612

Copyright Information

This work is made available under the terms of a Creative Commons Attribution License, available at https://creativecommons.org/licenses/by/4.0/

Peer reviewed



Geophysical Research Letters^{*}

RESEARCH LETTER

10.1029/2023GL104612

Key Points:

- Global forest carbon pools exhibit spatial heterogeneity driven by local differences in climatic and environmental factors
- Annual precipitation, elevation, soil, and wildfire were the primary controls of these differences in forest carbon pool fractions
- The ratio of biomass to ecosystem carbon in global forests will increase across the 21st century

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

Z. A. Mekonnen, zmekonnen@lbl.gov

Citation:

Mekonnen, Z. A., & Riley, W. J. (2023). Climate change will increase biomass proportion of global forest carbon stocks under an SSP5–8.5 climate trajectory. *Geophysical Research Letters*, 50, e2023GL104612. https://doi. org/10.1029/2023GL104612

Received 17 MAY 2023 Accepted 1 NOV 2023

© 2023 The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Climate Change Will Increase Biomass Proportion of Global Forest Carbon Stocks Under an SSP5–8.5 Climate Trajectory

Zelalem A. Mekonnen¹ D and William J. Riley¹ D

¹Climate and Ecosystem Sciences Division, Lawrence Berkeley National Laboratory, Berkeley, CA, USA

Abstract A large amount of carbon is stored in global forests. However, the fraction of carbon stored as plant biomass versus soil organic carbon (SOC) varies among forest types, and potential changes over the 21st century are uncertain. Here, we used extensive data derived from inventories and remote sensing and Coupled Model Intercomparison Project Phase 6 (CMIP6) models to examine the current and 21st century dynamics in the proportion of biomass and SOC across global forests. We found that precipitation, elevation, soil, and wildfire were the primary controls of these differences in carbon pools. Under the SSP5–8.5 climate scenario, CMIP6 models project that the ratio of biomass to ecosystem carbon in global forests will increase across the 21st century, with the largest increases in boreal forests ($95 \pm 37\%$) compared to moist tropical forests ($16 \pm 15\%$). Changes in forest carbon pools resulting in greater biomass fraction will affect disturbance, and ecosystem carbon and energy balances, all of which interact with the climate system.

Plain Language Summary The fraction of carbon stored as plant biomass versus soil organic carbon varies among forest types. The extent to which biomass and soil carbon pools may change over the 21st century is uncertain. In this study, we examined current and 21st century dynamics in the proportion of biomass and soil carbon across global forests. Our results show that global forest carbon pools exhibit spatial heterogeneity driven by local differences in climatic and environmental factors. Over the 21st century, we showed that the ratio of biomass to ecosystem carbon in global forests will increase, with the largest increases in boreal forests and lowest increases in moist tropical forests. These changes in forest ecosystem carbon pools have important implications for global ecosystem carbon balances and the climate system.

1. Introduction

Forests play an important role in the global carbon cycle by absorbing and storing a large amount of atmospheric CO_2 (Pan et al., 2011; Sedjo, 1993). Global forests contain more carbon in their biomass and soil than the total amount of carbon in the atmosphere (Pan et al., 2011). Over the past decades, global forests have been a strong carbon sink (Pan et al., 2011) with net ecosystem carbon uptake of 1.1 ± 0.8 Pg C yr⁻¹. Recent observations show that the contribution of forests in colder (e.g., boreal forest) versus warmer (e.g., tropical forest) climates differ in their contributions to the terrestrial carbon sink (Tagesson et al., 2020). Although tropical and boreal forest ecosystems store large amounts of total carbon, they have significant differences in their ecosystem structure and ability to store carbon in vegetation versus soil (Pan et al., 2011). With projected climate change, the extent to which global forest plant biomass and soil organic carbon (SOC) pools will change is uncertain (Ito et al., 2020; Koch et al., 2021). Climate change may affect these carbon pools through effects on plant carbon uptake and thus biomass accumulation, litter inputs to the soil, and SOC decomposition rates (Nottingham et al., 2015, 2020; L. Xu et al., 2021).

The boreal forest in the northern hemisphere high-latitude region contains large amounts of SOC that has accumulated over thousands of years (Hobbie et al., 2000; Oechel et al., 1992). Much of this SOC is trapped in permafrost (Natali et al., 2019; Schuur et al., 2015) as a result of colder climate and thus slower SOC decomposition (Gorham, 1991; Oechel et al., 1992). Boreal forest ecosystem carbon fixation rates are also slow from colder climate, nutrient limitations (Strömgren & Linder, 2002; Turkington et al., 1998), and a shorter growing season (Shirley et al., 2022; Suni et al., 2003), resulting in low plant productivity and thus biomass accumulation (Fernández-Martínez et al., 2014; Santoro et al., 2021). On the other hand, tropical forests in moist warmer climates have higher biomass driven by greater carbon fixation rates, and rapid SOC decomposition that enhances nutrient availability (Leff et al., 2012; Lewis et al., 2009; Vitousek & Sanford, 1986). Although litter inputs to the soil are high in moist tropical forests, rapid decomposition results in relatively high SOC turnover rates (Sayer et al., 2011; Stephenson & Mantgem, 2005). In temperate forests, soil parent material and litter quality have been shown to control forest soil carbon dynamics (Rasmussen et al., 2008). As a result of these variations in climatic and environmental drivers, global forests vary in ecosystem structure, spatial distribution, and relative amounts of accumulated carbon in plants and soil (Pan et al., 2011).

Anthropogenic climate warming and elevated atmospheric CO_2 concentrations are changing the carbon pools of global forest ecosystems (Lewis et al., 2009; Pan et al., 2011). Recent studies using measurements from ground, air, and space have shown increases in biomass across much of the global forests (Köhl et al., 2015; L. Xu et al., 2021). Changes in vegetation productivity inferred from Normalized Difference Vegetation Index trends, forest inventories, and tree-ring data across boreal forests show contrasting responses of greening, browning, and changes of biomass (Berner & Goetz, 2022; M. P. Girardin et al., 2016; Kauppi et al., 2010; Lloyd et al., 2011; Sulla-Menashe et al., 2018). Observed biomass increases in tropical forests have likely been in response to increasing atmospheric CO_2 concentrations (Lewis et al., 2009). Under a warmer climate, higher biomass of global forests (Köhl et al., 2015; L. Xu et al., 2021) may result in greater litter inputs to soil, which can increase SOC but also accelerate soil carbon release by stimulating existing carbon decomposition through priming (Sayer et al., 2011). In boreal regions, warming has threatened historical trends of SOC accumulation by driving rapid SOC decomposition from warmer soils and deepening of the active layer (Natali et al., 2019; Schuur et al., 2015). Effects of warming on SOC decomposition in tropical forests may vary depending on temperature sensitivity to decomposition and vegetation composition that controls litter quality (Nottingham et al., 2015; Yang & Chen, 2009).

In this study, we disentangled the current dominant climatic and environmental controls on the spatial distribution and relative proportion of biomass and SOC across global forests. We also examined changes to this proportion over the 21st century under climate change. We hypothesize that climate warming and increases in atmospheric CO_2 concentrations will increase the proportion of carbon stored as biomass, particularly in colder regions of the boreal and temperate versus tropical and subtropical forests.

2. Data and Methods

2.1. Study Area

Global forests cover 4.1×10^9 ha of land area (Dixon et al., 1994). In this study, we define the boundaries of global forest biomes based on the L. Xu et al. (2021) land cover map, which is derived from the MODIS International Geosphere-Biosphere Program land cover product. We used this vegetation classification to mask four major global forest biomes: moist tropical forests, tropical and subtropical dry forests, temperate forests, and boreal forests (Figure S1 in Supporting Information S1).

2.2. Biomass and SOC Stocks, Climatic, and Environmental Controls

Biomass data were obtained from a global above- and belowground live biomass carbon product (Saatchi et al., 2011), which was derived from in situ inventory plots, satellite light detection and ranging (Lidar) samples, and optical and microwave imagery (Table S1 in Supporting Information S1). Data for the total SOC stocks for global forests were taken from the Harmonized World Soil Database v1.2 data set (Wieder et al., 2014), which combined soil information from existing regional and national soil databases.

Many interacting factors affect biomass and SOC decomposition and accumulation in global forests. We identified eight major climatic and environmental controls (mean annual surface air temperature, annual precipitation, annual incoming shortwave radiation, elevation, parent material, wildfire, land cover change from forest harvest, and permafrost) that affect global forest productivity and thus biomass, soil development, and SOC accumulation (Table S1, Figure S2 in Supporting Information S1). As described below, we obtained these variables from global gridded data sets, which are derived from observations, inventories, and remote sensing products and models.

The surface air temperature data was taken from monthly CRU TS3.21 mean temperature observations (Jones & Harris, 2013). Precipitation data was generated from monthly CPC Merged Analysis of Precipitation global gridded precipitation product (Xie et al., 2007). The precipitation data was interpolated from the original resolution of $2.5^{\circ} \times 2.5^{\circ}$ to $0.5^{\circ} \times 0.5^{\circ}$. Surface incoming shortwave radiation was taken from Clouds and the Earth's Radiant Energy System Energy Balanced and Filled Top-of-Atmosphere (TOA) product (Loeb et al., 2018). The

original spatial resolution of $1^{\circ} \times 1^{\circ}$ was regridded to $0.5^{\circ} \times 0.5^{\circ}$. The elevation data were taken from the Global Multi-resolution Terrain Elevation Data 2010 (Danielson & Gesch, 2011). The soil parent material data were taken from Global Unconsolidated Sediments Map database (Börker et al., 2018). Total wildfire burned area fractions during fire events from 1997 to 2016 were derived from the Global Fire Emissions Database version 4.1 burned area fraction, which includes small fires (GFED4.1s) (Randerson et al., 2018). The total area of land cover change from forest harvest was derived from the Land-Use Harmonization 2 data set by merging total land cover change transitions among land-use states driven by wood harvest areas from primary, secondary mature, and secondary young forests (Hurtt et al., 2020). The permafrost (permanently frozen ground) spatial domain, which includes discontinuous, sporadic, and isolated permafrost boundaries, was taken from the Circum-Arctic Map of Permafrost and Ground-Ice Conditions, Version 2, published by National Snow and Ice Data Center (Brown et al., 2002). Based on this permafrost data, we re-grouped the data set and reclassified the global forest land grid cells into areas with versus without permafrost.

2.3. CMIP6 Models

Biomass and SOC stocks and thus their relative proportion across global forests may be altered under future climate. To examine the effects of projected climate warming and increased atmospheric CO_2 on global forest biomass and SOC stocks over the 21st century, we used modeled outputs of global forest biomass and SOC and calculated the ratio of biomass to ecosystem carbon (R_B) from 11 Coupled Model Intercomparison Project Phase 6 (CMIP6) models (ACCESS-ESM1-5, BCC-CSM2-MR, CESM2', CMCC-CM2-SR5, CNRM-ESM2-1, CanESM5, EC-Earth3, IPSL-CM6A-LR, NorESM2-LM, 'TaiESM1, 'UKESM1-0-LL) under the Shared Socio-economic Pathway (SSP5–8.5) climate scenario (Eyring et al., 2016). The SSP5–8.5 climate scenario was chosen since the trend of global carbon emissions is broadly consistent with this high emissions scenario. We also tested the performance of these CMIP6 models in simulating R_B against the data-derived values described above.

2.4. Data Analysis

To ensure consistent spatial resolution, all spatial data sets including biomass, SOC, climate and environmental drivers, and CMIP6 models were regridded to $0.5^{\circ} \times 0.5^{\circ}$ spatial resolution, across four major global forest biomes (moist tropical, tropical and subtropical dry, temperate, and boreal forests; Figure S1 in Supporting Information S1). We examined the relationship between climatic and environmental factors with global forest biomass, SOC, and R_B . The non-linear relationships and relative importance of the eight climatic and environmental factors that directly or indirectly affect plant biomass and SOC accumulation and thus control the spatial distribution of R_B across global forests were estimated using the random forest method (Breiman, 2001). We also examined ecosystem carbon turnover rates of global forests (Carvalhais et al., 2014) and the relative importance of soil properties (bulk density, field capacity, wilting point, thermal capacity, and total nitrogen density) to R_B using data (Figure S3 in Supporting Information S1) obtained from the Global Gridded Surfaces of Selected Soil Characteristics (Global, 2000).

The performance of CMIP6 models in simulating R_B under current climate was tested against data-derived values. These evaluations were conducted using a Taylor diagram (Taylor, 2001) that graphically illustrate the closeness of interannual and spatial patterns of the different sets of simulated biomass to ecosystem carbon ratios to the data-derived benchmark values. The Taylor diagram considered closeness of CMIP6 modeled versus data-derived values based on correlation, central root mean square difference (RMSD), and standard deviation (SD) between modeled and data-derived values. Simulated values nearest to the data-derived values were considered to be the closest to observations, as determined by highest correlation, lowest RMSD, and closest SD.

3. Results and Discussion

3.1. Spatial Distribution of Biomass and SOC Carbon Pools

Different global forest biomes exhibit unique patterns in the distribution of ecosystem carbon pools (Figure 1). Our results show contrasting biomass and SOC carbon pools across forests in colder versus moist warmer climates (Figure S4 in Supporting Information S1). Tropical forests have the largest spatial average biomass $(15 \pm 9 \text{ kgC m}^{-2})$ while the boreal forests have the smallest biomass $(2.6 \pm 2 \text{ kgC m}^{-2})$. In contrast, boreal





Figure 1. Observationally-inferred biomass to ecosystem carbon ratio varies across global forest biomes. Under current climate, data-derived plant biomass to ecosystem carbon ratios ((plant biomass/ecosystem carbon) $\times 100$; ecosystem carbon = plant biomass + soil organic carbon) across (a) global land areas covered by forests, (b) global forest biomes (moist tropical forests (MTF), tropical and subtropical dry forests (TSDF), temperate forests (TF), and boreal forests (BF); Figure S1 in Supporting Information S1), and (c) across latitudes. The error bars in panel (b) and the shaded area in panel (c) represent one standard deviation.

forests store the largest amount of SOC ($19 \pm 8 \text{ kgC m}^{-2}$) followed by temperate forests ($13.6 \pm 5 \text{ kgC m}^{-2}$). The temperate and tropical and subtropical dry forests have 11.6 ± 4 and $9.5 \pm 4 \text{ kgC m}^{-2}$, respectively. We found that the distribution in the R_B of global forests has distinct latitudinal gradients (Figure 1). R_B is highest in tropical moist climates and decreases northward and southward from the tropics (Figure S4 in Supporting Information S1). Most boreal forest ecosystem carbon is stored as SOC with only $14 \pm 7\%$ in plant biomass (Figure 1b). In contrast, moist tropical forests store $50 \pm 18\%$ of their carbon as biomass. The temperate and tropical and subtropical dry forests have intermediate values of biomass carbon ratios ($28 \pm 13\%$ and $21 \pm 12\%$, respectively). These results are consistent with other studies that demonstrate variations in plant and soil carbon pools among different forests (Eswaran et al., 1993; Post et al., 1982). While colder regions exhibit lower biomass carbon ratios compared to the warmer moist tropics, our results show substantial spatial heterogeneity within each forest biome; for example, the high ratios in parts of the Eurasian boreal forests and the Amazon and Congo Basin moist tropical forests.

3.2. Climatic and Environmental Controls of Forest Carbon

Forest biomass and SOC pools are controlled by several climatic and environmental factors. Our results from the random forest model using data derived from inventories and remote sensing ($R^2 = 0.63$) show that annual precipitation, elevation, soil, and wildfire are the most important environmental and climatic controls on R_B (Figure 2). These controls differ in their relative effects on R_B , biomass, and SOC pools of global forests (Figure 2; Figure S3 in Supporting Information S1). For example, annual precipitation is the most important controller for R_B (Figure 2) through effects on both biomass and SOC stocks (Figure S5 in Supporting Information S1). The



Figure 2. Annual precipitation is the most important climatic control on observed biomass to ecosystem carbon ratio. The relative importance of the eight climatic and environmental factors that affect observed biomass to ecosystem carbon ratio estimated using a random forest method (Breiman, 2001): (a) unbiased predictor importance estimates and (b) predictor association estimates. T air denotes surface air temperature, and SW denotes incoming shortwave radiation.

importance of precipitation in controlling forest carbon pools is consistent with a study that shows soil and biomass carbon pools across precipitation gradients (Meier & Leuschner, 2010). Elevation, primarily through its effects on SOC, is the second most important factor that affects R_B (Figure 2; Figure S3 in Supporting Information S1). Elevation indirectly affects several more direct controllers on R_B , including air temperature, precipitation, and soil development (Moser et al., 2011). Variations in landscape elevation can indirectly influence plant growth and the subsequent accumulation of soil carbon through its effect on hydrology. Soil parent material affects soil formation and subsequent soil physical and chemical properties (Wilson, 2019), consistent with its large inferred control on SOC stocks (Figure S5 in Supporting Information S1). Wildfires also affect forest biomass and SOC (Curtis et al., 2018; Turetsky et al., 2010; Van Der Werf et al., 2003), and recent estimates from satellite-derived data show that $38 \pm 9\%$ of global forest loss was associated with fire (van Wees et al., 2021). Consistent with that study, our results show that wildfire is one of the dominant controllers of global forest biomass and SOC (Figure S5 in Supporting Information S1). We also conducted further analysis to examine the effects of soil physical, hydrological, and thermal properties and identified soil nitrogen content, field capacity, and wilting point as important soil variables that control R_B (Figure S6 in Supporting Information S1). These soil properties affect soil moisture and nutrients availability and thus forest growth and biomass accumulation (Kimmins et al., 1990; Pastor & Post, 1986).

3.3. Forest Carbon Predictions of CMIP6 Models

Our analysis using 11 CMIP6 models shows that prediction of current global forest carbon pools vary widely among the models (Figure 3; Figure S7 in Supporting Information S1). Spatially averaged across latitudes, most of the models did not accurately capture the data-derived R_B patterns (Figure 3a; Figure S8 in Supporting Information S1). Large differences also existed in correlations, RMSDs, and SD between modeled and data-derived values (Figure 3b). However, consistent with data-derived values, the multimodel ensemble spatial average R_B were lower in boreal and temperate versus in moist tropical forests. R_B values were overestimated in tropical and subtropical dry forests (Figure 3c). In contrast to the observations, the models exhibit greater latitudinal variability in the northern hemisphere compared to the southern hemisphere. While multimodel mean R_B values are generally consistent with data-derived values, in some regions, particularly within 20°N–30°N, the multimodel mean substantially overestimated R_B (Figure 3a). This bias was partly attributed to overestimation of modeled biomass in the tropical and subtropical dry forests and underestimation of biomass in parts of the moist tropical forests (Figure S7 in Supporting Information S1). Greater differences in CMIP6 modeled SOC were found more in the boreal than tropical forests (Figure S7 in Supporting Information S1). While the spatially averaged SOC values were reasonably captured by the multimodel mean across three of the four global forest biomes, most



Geophysical Research Letters



Figure 3. Coupled Model Intercomparison Project Phase 6 (CMIP6) models predict very different biomass to ecosystem carbon ratios. Eleven CMIP6 models predicted ratios of biomass to ecosystem carbon (a) zonally compared with data-derived values; (b) differences in correlations, root-mean square differences, and standard deviations between modeled and data-derived values; and (c) multimodel mean spatially averages across global forest biomes. The shaded region in panel (a) represents one standard deviation computed from 11 CMIP6 models, and the error bars in panel (b) represent one standard deviation computed from spatial variability.

models and the multimodel mean overestimated SOC in the boreal forest regions of the northern hemisphere (Figure S7 in Supporting Information S1).

3.4. 21st Century Changes in Fraction of Forest Biomass Carbon

Our analysis of CMIP6 models under the SSP5–8.5 climate scenario shows that biomass will increase across much of the global forests in the 21st century (Figure 4a). However, localized declines in biomass were modeled in parts of the Amazon and the Congo basin. Changes in SOC exhibit spatial heterogeneity with overall declines in the boreal forests and both increases and decreases in parts of the tropical forests (Figure 4b). Observational studies have indicated that increases in biomass likely results in increased litter inputs to the soil, thereby accelerating SOC decomposition in high-latitude regions under warming (Bronson et al., 2008; Lawrence & Slater, 2005; S. Xu et al., 2013), and accelerating decomposition in the tropics stimulated by fresh litter inputs (Sayer et al., 2011, 2019). Although these mechanisms are represented differently (or not at all) in the CMIP6 models, the models project SOC declines or slight increases across much of the global forests (Figures 4a and 4b).

Projected changes in multimodal mean R_B values of the 11 CMIP6 models had distinct patterns across the global forests (Figure 4c). These results show that global forests will gain biomass carbon and lose SOC in much of the forest regions by the year 2100. Projected 21st century changes from these CMIP6 models show that R_B will



Geophysical Research Letters



Figure 4. Projected changes in biomass, soil organic carbon (SOC), and ratio of biomass to ecosystem carbon vary across forests in the 21st century. The Coupled Model Intercomparison Project Phase 6 multimodel mean changes (2100–2015) in (a) biomass carbon, (b) SOC, (c) % change in R_B values, and (d) spatial average % change in R_B values across the four major global forest biomes. The error bars in panel (d) represent one standard deviation.

increase across much of the global forests (Figure 4). These increases are more pronounced in the boreal and temperate forests versus a slight increase or localized declines in tropical forests (particularly parts of the Amazon and Congo basin) (Figure 4a). The multimodel spatial average R_B for the boreal forests will increase by $95 \pm 37\%$ compared to only a $16 \pm 15\%$ increase for moist tropical forests by the year 2100 (Figure 4b). Intermediate increases in spatial average R_B values were found for temperate forests (22%) and tropical and subtropical dry forests (39%)). These differences in R_B changes are partly attributed to the higher positive temperature sensitivity of carbon fixation to climate warming that enhances productivity (Kauppi et al., 2014) and drives higher forest biomass (Kauppi et al., 2010; Stinziano & Way, 2014) in the colder boreal and temperate climates (Figure 4), as opposed to adverse effects of warming in tropical climates (Doughty & Goulden, 2008). In the tropical climate, the adverse effect of warming is partly offset by increased atmospheric CO₂ concentrations that benefit forest growth through enhancing carbon fixation (Fernández-Martínez et al., 2018) and increasing water use efficiency (Keenan et al., 2013). Similarly, warming may affect modeled SOC dynamics through its effect on decomposition rates and indirectly through alterations in plant functional types and biomass, which, in turn, modify litter quantity and quality (Conant et al., 2008; Hopkins et al., 2012; Wang et al., 2016).

Our results highlight that global forests have very different relative proportions of plant biomass and SOC pools. While spatial heterogeneity in carbon pools is driven by local variations in climatic and environmental factors, the proportion of forest ecosystem carbon in plants versus soil follows distinct latitudinal gradients and varies among different forests (Figure 1). The R_B derived from data and CMIP6 models are also related to forest productivity and carbon turnover rates. We found from data-derived products that under current conditions, ecosystem carbon turnover rates vary with forest biomes (Figure S9 in Supporting Information S1). This variation in turnover time spatial pattern is inversely related to variation in biomass and R_B (Figure 1 vs. Figure S9 in Supporting Information S1) and is consistent with studies that concluded that global forest turnover rates are linked with forest net primary productivity, and thus biomass (C. A. J. Girardin et al., 2010; Keeling & Phillips, 2007; Stephenson & Mantgem, 2005).

We note that our estimates of the relative magnitude of biomass and SOC pools under current and 21st century conditions may be subject to a number of uncertainties. For instance, the spatial resolution of the original data

sets we used were different and sub-gridcell heterogeneity below our $0.5^{\circ} \times 0.5^{\circ}$ spatial resolution is not captured in our analysis and should be addressed in future study. Uncertainties in modeled soil carbon stocks attributed to differences in belowground process representations were also shown in CMIP6 models (Varney et al., 2022). Fan et al. (2020) also reported that uncertainties in ecosystem carbon turnover rates are mostly attributed to global soil carbon estimates. Our results are also subject to uncertainties from CMIP6 models driven by differences in model structure and parameterization, vegetation types, and climate (Eyring et al., 2016). In this regard, recent work has argued that model selection for ensemble calculations should account for model accuracy under current conditions, rather than assuming a "model democracy" (Hausfather et al., 2022). We therefore re-evaluated our conclusions using a "model meritocracy," that is, the 4 best CMIP6 models compared to current observations (Figure S10 in Supporting Information S1). We found no substantial differences in changes in spatial pattern and relative proportions of plant biomass to ecosystem carbon ratio of the global forests compared to results from the overall ensemble, lending additional confidence to our results.

4. Conclusions

We found from data-derived products and CMIP6 models that global forests have very different structure and ability to store carbon in vegetation versus soil. Under current conditions, these differences in ecosystem carbon pools are mainly controlled by annual precipitation, elevation, soil parent material, and wildfire (Figure 2). Warmer 21st century climate and wildfire will likely increase the relative proportion of plant biomass carbon in ecosystems (Figure 4). Our result of projected increases in biomass versus SOC under 21st century climate may result in several ecological and climatic feedbacks. Forest ecosystems with greater biomass typically have higher fuel loads that can exacerbate wildfire through interaction with warmer climates (Pausas & Keeley, 2021). Biomass is more vulnerable than SOC to natural disturbances, such as fungal and insect attacks, drought, and forest dieback that reduce ecosystem carbon uptake (Allen et al., 2010; Doughty et al., 2015; Huang & Anderegg, 2012), and these disturbances are not well-represented in CMIP6 models. Biomass is also vulnerable to anthropogenic disturbances such as deforestation. Overall, alterations in the structure of global forests and the distribution of carbon within ecosystems, leading to a higher proportion of carbon stored in plant biomass, will have implications for surface energy balances, disturbance regimes, ecosystem carbon balance, and thus feedbacks with the climate system.

Data Availability Statement

Data products that are used and support the findings of this study are publicly available at https://www.ilamb.org/ datasets.html and https://esgf-node.llnl.gov/projects/cmip6/.

References

- Allen, C. D., Macalady, A. K., Chenchouni, H., Bachelet, D., McDowell, N., Vennetier, M., et al. (2010). A global overview of drought and heat-induced tree mortality reveals emerging climate change risks for forests. Forest Ecology and Management, 259(4), 660-684. https://doi. org/10.1016/i.foreco.2009.09.001
- Berner, L. T., & Goetz, S. J. (2022). Satellite observations document trends consistent with a boreal forest biome shift. Global Change Biology, 28(10), 3275-3292. https://doi.org/10.1111/gcb.16121
- Börker, J., Hartmann, J., Amann, T., & Romero-Mujalli, G. (2018). Terrestrial sediments of the Earth: Development of a global unconsolidated sediments map database (GUM). Geochemistry, Geophysics. Geosystems, 19(4), 997-1024. https://doi.org/10.1002/2017gc007273

Breiman, L. (2001). Random forests. Machine Learning, 45(1), 5-32. https://doi.org/10.1023/a:1010933404324

- Bronson, D. R., Gower, S. T., Tanner, M., Linder, S., & Van Herk, I. (2008). Response of soil surface CO, flux in a boreal forest to ecosystem warming. Global Change Biology, 14(4), 856-867. https://doi.org/10.1111/j.1365-2486.2007.01508.x
- Brown, J., Ferrians, O., Heginbottom, J. A., & Melnikov, A. E. (2002). Circum-Arctic map of permafrost and ground-ice conditions, version 2 [Dataset]. National Snow and Ice Data Center. https://doi.org/10.7265/skbg-kf16

Carvalhais, N., Forkel, M., Khomik, M., Bellarby, J., Jung, M., Migliavacca, M., et al. (2014). Global covariation of carbon turnover times with climate in terrestrial ecosystems. Nature, 514(7521), 213-217. https://doi.org/10.1038/nature13731

- Conant, R. T., Drijber, R. A., Haddix, M. L., Parton, W. J., Paul, E. A., Plante, A. F., et al. (2008). Sensitivity of organic matter decomposition to warming varies with its quality. Global Change Biology, 14(4), 868-877. https://doi.org/10.1111/j.1365-2486.2008.01541.x
- Curtis, P. G., Slay, C. M., Harris, N. L., Tyukavina, A., & Hansen, M. C. (2018). Classifying drivers of global forest loss. Science, 361(6407), 1108-1111. https://doi.org/10.1126/science.aau3445
- Danielson, J. J., & Gesch, D. B. (2011). Global multi-resolution terrain elevation data 2010 (GMTED2010). Retrieved from https://pubs.usgs.gov/of/2011/1073/

Dixon, R. K., Solomon, A. M., Brown, S., Houghton, R. A., Trexier, M. C., & Wisniewski, J. (1994). Carbon pools and flux of global forest ecosystems. Science, 263(5144), 185-190. https://doi.org/10.1126/science.263.5144.185

Acknowledgments

This research was supported by the Director, Office of Science, Office of Biological and Environmental Research, US Department of Energy through the Regional and Global Model Analysis (RGMA) Program-Reducing Uncertainty in Biogeochemical Interactions through Synthesis and Computation (RUBISCO) Scientific Focus Area (SFA). Lawrence Berkeley National Laboratory (LBNL) is managed by the University of California for the U.S. Department of Energy under contract no. DEAC02-05CH11231

articles

are governed by the applicable Creative Commons Licens.

19448007, 2023, 23, Downloaded from https://agupubs.onlinelibrary.viley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL104612, Wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.wiley Online Library on [30/04/2024]. See the Terms and Conditions (https://onlinelibrary.

- Doughty, C. E., & Goulden, M. L. (2008). Are tropical forests near a high temperature threshold? *Journal of Geophysical Research*, 113(G1), G00B07. https://doi.org/10.1029/2007jg000632
- Doughty, C. E., Metcalfe, D. B., Girardin, C. A. J., Amézquita, F. F., Cabrera, D. G., Huasco, W. H., et al. (2015). Drought impact on forest carbon dynamics and fluxes in Amazonia. *Nature*, 519(7541), 78–82. https://doi.org/10.1038/nature14213
- Eswaran, H., Van Den Berg, E., & Reich, P. (1993). Organic carbon in soils of the world. Soil Science Society of America, 57(1), 192–194. https:// doi.org/10.2136/sssaj1993.03615995005700010034x
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the coupled model Intercomparison project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Fan, N., Koirala, S., Reichstein, M., Thurner, M., Avitabile, V., Santoro, M., et al. (2020). Apparent ecosystem carbon turnover time: Uncertainties and robust features. *Earth System Science Data*, 12(4), 2517–2536. https://doi.org/10.5194/essd-12-2517-2020
- Fernández-Martínez, M., Sardans, J., Chevallier, F., Ciais, P., Obersteiner, M., Vicca, S., et al. (2018). Global trends in carbon sinks and their relationships with CO₂ and temperature. *Nature Climate Change*, 9(1), 73–79. https://doi.org/10.1038/s41558-018-0367-7
- Fernández-Martínez, M., Vicca, S., Janssens, I. A., Sardans, J., Luyssaert, S., Campioli, M., et al. (2014). Nutrient availability as the key regulator of global forest carbon balance. *Nature Climate Change*, 4(6), 471–476. https://doi.org/10.1038/nclimate2177
- Girardin, C. A. J., Malhi, Y., Aragão, L. E. O. C., Mamani, M., Huaraca Huasco, W., Durand, L., et al. (2010). Net primary productivity allocation and cycling of carbon along a tropical forest elevational transect in the Peruvian Andes. *Global Change Biology*, 16(12), 3176–3192. https:// doi.org/10.1111/j.1365-2486.2010.02235.x
- Girardin, M. P., Bouriaud, O., Hogg, E. H., Kurz, W., Zimmermann, N. E., Metsaranta, J. M., et al. (2016). No growth stimulation of Canada's boreal forest under half-century of combined warming and CO2 fertilization. *Proceedings of the National Academy of Sciences of the United States of America*, 113(52), E8406–E8414. https://doi.org/10.1073/pnas.1610156113
- Global, S. D. T. (2000). Global gridded surfaces of selected soil characteristics (igbp-dis) [Dataset]. ORNL Distributed Active Archive Center Datasets. ORNL Environmental Sciences Division. https://doi.org/10.3334/ornldaac/569
- Gorham, E. (1991). Northern peatlands: Role in the carbon cycle and probable responses to climatic warming. *Ecological Applications: A Publication of the Ecological Society of America*, 1(2), 182–195. https://doi.org/10.2307/1941811
- Hausfather, Z., Marvel, K., Schmidt, G. A., Nielsen-Gammon, J. W., & Zelinka, M. (2022). Climate simulations: Recognize the "hot model" problem. *Nature*, 605(7908), 26–29. https://doi.org/10.1038/d41586-022-01192-2
- Hobbie, S. E., Schimel, J. P., Trumbore, S. E., & Randerson, J. R. (2000). Controls over carbon storage and turnover in high-latitude soils. *Global Change Biology*, 6(S1), 196–210. https://doi.org/10.1046/j.1365-2486.2000.06021.x
- Hopkins, F. M., Torn, M. S., & Trumbore, S. E. (2012). Warming accelerates decomposition of decades-old carbon in forest soils. Proceedings of the National Academy of Sciences of the United States of America, 109(26), E1753–E1761. https://doi.org/10.1073/pnas.1120603109
- Huang, C.-Y., & Anderegg, W. R. L. (2012). Large drought-induced aboveground live biomass losses in southern Rocky Mountain aspen forests. *Global Change Biology*, 18(3), 1016–1027. https://doi.org/10.1111/j.1365-2486.2011.02592.x
- Hurtt, G. C., Chini, L., Sahajpal, R., Frolking, S., Bodirsky, B. L., Calvin, K., et al. (2020). Harmonization of global land use change and management for the period 850–2100 (LUH2) for CMIP6. *Geoscientific Model Development*, 13(11), 5425–5464. https://doi.org/10.5194/ gmd-13-5425-2020
- Ito, A., Hajima, T., Lawrence, D. M., Brovkin, V., Delire, C., Guenet, B., et al. (2020). Soil carbon sequestration simulated in CMIP6-LUMIP models: Implications for climatic mitigation. *Environmental Research Letters: ERL*, 15(12), 124061. https://doi.org/10.1088/1748-9326/ abc912
- Jones, P. D., & Harris, I. C. (2013). High resolution gridded data of month-by-month variation in climate. NCAS British Atmospheric Data Centre, Climatic Research Unit (CRU) Time-Series (TS) Version 3.21. https://doi.org/10.5285/D0E1585D-3417-485F-87AE-4FCECF10A992
- Kauppi, P. E., Posch, M., & Pirinen, P. (2014). Large impacts of climatic warming on growth of boreal forests since 1960. *PLoS One*, 9(11), e111340. https://doi.org/10.1371/journal.pone.0111340
- Kauppi, P. E., Rautiainen, A., Korhonen, K. T., Lehtonen, A., Liski, J., Nöjd, P., et al. (2010). Changing stock of biomass carbon in a boreal forest over 93 years. Forest Ecology and Management, 259(7), 1239–1244. https://doi.org/10.1016/j.foreco.2009.07.044
- Keeling, H. C., & Phillips, O. L. (2007). The global relationship between forest productivity and biomass. Global Ecology and Biogeography: A Journal of Macroecology, 16(5), 618–631. https://doi.org/10.1111/j.1466-8238.2007.00314.x
- Keenan, T. F., Hollinger, D. Y., Bohrer, G., Dragoni, D., Munger, J. W., Schmid, H. P., & Richardson, A. D. (2013). Increase in forest water-use efficiency as atmospheric carbon dioxide concentrations rise. *Nature*, 499(7458), 324–327. https://doi.org/10.1038/nature12291
- Kimmins, J. P., Comeau, P. G., & Kurz, W. (1990). Modelling the interactions between moisture and nutrients in the control of forest growth. Forest Ecology and Management. 30(1), 361–379. https://doi.org/10.1016/0378-1127(90)90149-6
- Soch, A., Hubau, W., & Lewis, S. L. (2021). Earth system models are not capturing present-day tropical forest carbon dynamics. *Earth's Future*, 9(5), e2020EF001874. https://doi.org/10.1029/2020ef001874
- Köhl, M., Lasco, R., Cifuentes, M., Jonsson, Ö., Korhonen, K. T., Mundhenk, P., et al. (2015). Changes in forest production, biomass and carbon: Results from the 2015 UN FAO global forest resource assessment. *Forest Ecology and Management*, 352, 21–34. https://doi.org/10.1016/j. foreco.2015.05.036
- Lawrence, D. M., & Slater, A. G. (2005). A projection of severe near-surface permafrost degradation during the 21st century. *Geophysical Research Letters*, 32(24), L24401. https://doi.org/10.1029/2005gl025080
- Leff, J. W., Wieder, W. R., Taylor, P. G., Townsend, A. R., Nemergut, D. R., Grandy, A. S., & Cleveland, C. C. (2012). Experimental litterfall manipulation drives large and rapid changes in soil carbon cycling in a wet tropical forest. *Global Change Biology*, 18(9), 2969–2979. https:// doi.org/10.1111/j.1365-2486.2012.02749.x
- Lewis, S. L., Lloyd, J., Sitch, S., Mitchard, E. T. A., & Laurance, W. F. (2009). Changing ecology of tropical forests: Evidence and drivers. *Annual Review of Ecology, Evolution and Systematics*, 40(1), 529–549. https://doi.org/10.1146/annurev.ecolsys.39.110707.173345
- Lloyd, A. H., Bunn, A. G., & Berner, L. (2011). A latitudinal gradient in tree growth response to climate warming in the Siberian taiga. Global Change Biology, 17(5), 1935–1945. https://doi.org/10.1111/j.1365-2486.2010.02360.x
- Loeb, N. G., Doelling, D. R., Wang, H., Su, W., Nguyen, C., Corbett, J. G., et al. (2018). Clouds and the Earth's radiant energy system (CERES) energy balanced and filled (EBAF) top-of-atmosphere (TOA) edition-4.0 data product. *Journal of Climate*, *31*(2), 895–918. https://doi.org/10.1175/jcli-d-17-0208.1
- Meier, I. C., & Leuschner, C. (2010). Variation of soil and biomass carbon pools in beech forests across a precipitation gradient. *Global Change Biology*, 16(3), 1035–1045. https://doi.org/10.1111/j.1365-2486.2009.02074.x

- Moser, G., Leuschner, C., Hertel, D., Graefe, S., Soethe, N., & Iost, S. (2011). Elevation effects on the carbon budget of tropical mountain forests (S Ecuador): The role of the belowground compartment. *Global Change Biology*, *17*(6), 2211–2226. https://doi.org/10.1111/j.1365-2486.2010.02367.x
- Natali, S. M., Watts, J. D., Rogers, B. M., Potter, S., Ludwig, S. M., Selbmann, A.-K., et al. (2019). Large loss of CO₂ in winter observed across the northern permafrost region. *Nature Climate Change*, 9(11), 852–857. https://doi.org/10.1038/s41558-019-0592-8

Nottingham, A. T., Meir, P., Velasquez, E., & Turner, B. L. (2020). Soil carbon loss by experimental warming in a tropical forest. *Nature*, 584(7820), 234–237. https://doi.org/10.1038/s41586-020-2566-4

- Nottingham, A. T., Whitaker, J., Turner, B. L., Salinas, N., Zimmermann, M., Malhi, Y., & Meir, P. (2015). Climate warming and soil carbon in tropical forests: Insights from an elevation gradient in the Peruvian Andes. *BioScience*, 65(9), 906–921. https://doi.org/10.1093/biosci/biv109
- Oechel, W. C., Billings, W. D., et al. (1992). Effects of global change on the carbon balance of arctic plants and ecosystems. Arctic Ecosystems in a Changing Climate: An Ecophysiological Perspective, 139–168.
- Pan, Y., Birdsey, R. A., Fang, J., Houghton, R., Kauppi, P. E., Kurz, W. A., et al. (2011). A large and persistent carbon sink in the world's forests. Science, 333(6045), 988–993. https://doi.org/10.1126/science.1201609
- Pastor, J., & Post, W. M. (1986). Influence of climate, soil moisture, and succession on forest carbon and nitrogen cycles. *Biogeochemistry*, 2(1), 3–27. https://doi.org/10.1007/bf02186962
- Pausas, J. G., & Keeley, J. E. (2021). Wildfires and global change. Frontiers in Ecology and the Environment, 19(7), 387–395. https://doi. org/10.1002/fee.2359
- Post, W. M., Emanuel, W. R., Zinke, P. J., & Stangenberger, A. G. (1982). Soil carbon pools and world life zones. Nature, 298(5870), 156–159. https://doi.org/10.1038/298156a0
- Randerson, J. T., Van Der Werf, G. R., Giglio, L., Collatz, G. J., & Kasibhatla, P. S. (2018). Global fire emissions database, version 4, (GFEDv4). ORNL DAAC. 0.3334/ORNLDAAC/1293. Retrieved from http://daac.ornl.gov/VEGETATION/guides/fire_emissions_v4.html
- Rasmussen, C., Southard, R. J., & Horwath, W. R. (2008). Litter type and soil minerals control temperate forest soil carbon response to climate change. *Global Change Biology*, *14*(9), 2064–2080. https://doi.org/10.1111/j.1365-2486.2008.01639.x
- Saatchi, S. S., Harris, N. L., Brown, S., Lefsky, M., Mitchard, E. T. A., Salas, W., et al. (2011). Benchmark map of forest carbon stocks in tropical regions across three continents. Proceedings of the National Academy of Sciences of the United States of America, 108(24), 9899–9904. https:// doi.org/10.1073/pnas.1019576108
- Santoro, M., Cartus, O., Carvalhais, N., Rozendaal, D. M. A., Avitabile, V., Araza, A., et al. (2021). The global forest above-ground biomass pool for 2010 estimated from high-resolution satellite observations. *Earth System Science Data*, 13(8), 3927–3950. https://doi.org/10.5194/ essd-13-3927-2021
- Sayer, E. J., Heard, M. S., Grant, H. K., Marthews, T. R., & Tanner, E. V. J. (2011). Soil carbon release enhanced by increased tropical forest litterfall. *Nature Climate Change*, 1(6), 304–307. https://doi.org/10.1038/nclimate1190
- Sayer, E. J., Lopez-Sangil, L., Crawford, J. A., Bréchet, L. M., Birkett, A. J., Baxendale, C., et al. (2019). Tropical forest soil carbon stocks do not increase despite 15 years of doubled litter inputs. *Scientific Reports*, 9(1), 18030. https://doi.org/10.1038/s41598-019-54487-2
- Schuur, E. A. G., McGuire, A. D., Schädel, C., Grosse, G., Harden, J. W., Hayes, D. J., et al. (2015). Climate change and the permafrost carbon feedback. *Nature*, 520(7546), 171–179. https://doi.org/10.1038/nature14338
- Sedjo, R. A. (1993). The carbon cycle and global forest ecosystem. Terrestrial Biospheric Carbon Fluxes, 295-307. https://doi.org/10.1007/978-94-011-1982-5_19
- Shirley, I. A., Mekonnen, Z. A., Grant, R. F., Dafflon, B., Hubbard, S. S., & Riley, W. J. (2022). Rapidly changing high-latitude seasonality: Implications for the 21st century carbon cycle in Alaska. *Environmental Research Letters*, 17(1), 014032, https://doi.org/10.1088/1748-9326/ac4362
- Stephenson, N. L., & Mantgem, P. J. (2005). Forest turnover rates follow global and regional patterns of productivity. *Ecology Letters*, 8(5), 524–531. https://doi.org/10.1111/j.1461-0248.2005.00746.x
- Stinziano, J. R., & Way, D. A. (2014). Combined effects of rising [CO₂] and temperature on boreal forests: Growth, physiology and limitations. *Botany*, 92(6), 425–436. https://doi.org/10.1139/cjb-2013-0314
- Strömgren, M., & Linder, S. (2002). Effects of nutrition and soil warming on stemwood production in a boreal Norway spruce stand. Global Change Biology, 8(12), 1194–1204. https://doi.org/10.1046/j.1365-2486.2002.00546.x
- Sulla-Menashe, D., Woodcock, C. E., & Friedl, M. A. (2018). Canadian boreal forest greening and browning trends: An analysis of biogeographic patterns and the relative roles of disturbance versus climate drivers. *Environmental Research Letters: ERL*, 13(1), 014007. https://doi. org/10.1088/1748-9326/aa9b88
- Suni, T., Berninger, F., Markkanen, T., Keronen, P., Rannik, Ü., & Vesala, T. (2003). Interannual variability and timing of growing-season CO₂ exchange in a boreal forest. *Journal of Geophysical Research*, 108(D9), . https://doi.org/10.1029/2002jd002381
- Tagesson, T., Schurgers, G., Horion, S., Ciais, P., Tian, F., Brandt, M., et al. (2020). Recent divergence in the contributions of tropical and boreal forests to the terrestrial carbon sink. *Nature Ecology & Evolution*, 4(2), 202–209. https://doi.org/10.1038/s41559-019-1090-0
- Taylor, K. E. (2001). Summarizing multiple aspects of model performance in a single diagram. *Journal of Geophysical Research*, 106(D7), 7183–7192. https://doi.org/10.1029/2000jd900719
- Turetsky, M. R., Kane, E. S., Harden, J. W., Ottmar, R. D., Manies, K. L., Hoy, E., & Kasischke, E. S. (2010). Recent acceleration of biomass burning and carbon losses in Alaskan forests and peatlands. *Nature Geoscience*, 4(1), 27–31. https://doi.org/10.1038/ngeo1027

Turkington, R., John, E., Krebs, C. J., Dale, M. R. T., Nams, V. O., Boonstra, R., et al. (1998). The effects of NPK fertilization for nine years on boreal forest vegetation in northwestern Canada. *Journal of Vegetation Science*, 9(3), 333–346. https://doi.org/10.2307/3237098

- Van Der Werf, G. R., Randerson, J. T., Collatz, G. J., & Giglio, L. (2003). Carbon emissions from fires in tropical and subtropical ecosystems. *Global Change Biology*, 9(4), 547–562. https://doi.org/10.1046/j.1365-2486.2003.00604.x
- van Wees, D., van der Werf, G. R., Randerson, J. T., Andela, N., Chen, Y., & Morton, D. C. (2021). The role of fire in global forest loss dynamics. Global Change Biology, 27(11), 2377–2391. https://doi.org/10.1111/gcb.15591

Varney, R. M., Chadburn, S. E., Burke, E. J., & Cox, P. M. (2022). Evaluation of soil carbon simulation in CMIP6 Earth system models. *Biogeosciences*, 19(19), 4671–4704. https://doi.org/10.5194/bg-19-4671-2022

Vitousek, P. M., & Sanford, R. L., Jr. (1986). Nutrient cycling in moist tropical forest. Annual Review of Ecology and Systematics, 17(1), 137–167. https://doi.org/10.1146/annurev.es.17.110186.001033

- Wang, Q., He, T., & Liu, J. (2016). Litter input decreased the response of soil organic matter decomposition to warming in two subtropical forest soils. Scientific Reports, 6(1), 33814. https://doi.org/10.1038/srep33814
- Wieder, W. R., Boehnert, J., Bonan, G. B., & Langseth, M. (2014). Regridded harmonized world soil database v1. 2. ORNL DAAC. Retrieved from https://daac.ornl.gov/cgi-bin/download.pl?ds_id=1247&source=schema_org_metadata
- Wilson, M. J. (2019). The importance of parent material in soil classification: A review in a historical context. *Catena*, 182, 104131. https://doi.org/10.1016/j.catena.2019.104131

- Xie, P., Arkin, P. A., & Janowiak, J. E. (2007). CMAP: The CPC merged analysis of precipitation. In V. Levizzani, P. Bauer, & F. J. Turk (Eds.), *Measuring precipitation from space: EURAINSAT and the future* (pp. 319–328). Springer.
- Xu, L., Saatchi, S. S., Yang, Y., Yu, Y., Pongratz, J., Bloom, A. A., et al. (2021). Changes in global terrestrial live biomass over the 21st century. Science Advances, 7(27), eabe9829. https://doi.org/10.1126/sciadv.abe9829
- Xu, S., Liu, L. L., & Sayer, E. J. (2013). Variability of above-ground litter inputs alters soil physicochemical and biological processes: A meta-analysis of litterfall-manipulation experiments. *Biogeosciences*, 10(11), 7423–7433. https://doi.org/10.5194/bg-10-7423-2013
- Yang, X., & Chen, J. (2009). Plant litter quality influences the contribution of soil fauna to litter decomposition in humid tropical forests, southwestern China. Soil Biology and Biochemistry, 41(5), 910–918. https://doi.org/10.1016/j.soilbio.2008.12.028